Author’s response to reviews

Title: Continuous Low-Intensity Ultrasound Attenuates IL-6 and TNFα-induced Catabolic Effects and Repairs Chondral Fissures in Bovine Osteochondral Explants

Authors:

Neety Sahu (neety.sahu@huskers.unl.edu)

Hendrik Viljoen (hviljoen1@unl.edu)

Anuradha Subramanian (anu.subramanian@uah.edu)

Version: 3 Date: 08 Apr 2019

Author’s response to reviews:

Thank you for the peer-review of our manuscript. We appreciate the comments and recommendations made by the review process. Per your suggestions, we have addressed the minor revisions.

Response to Editor Comments:

1. Authors’ Contributions:

We have noted that author Hendrik J. Viljoen is missing in the listed authors’ contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship

Response: Thank you for identifying the error. We have included Dr. Hendrik J. Viljoen’s (HV) contribution.

2. Supplementary Figure 2

Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. We note that Supplementary Figure 2 is not currently cited within the text.
Response: Thank you for the comment. We have removed Supplementary Figure 2 from the manuscript.

3. Clean manuscript:

Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: We have provided a clean manuscript in accordance with the indicated guidelines.